Author's response to reviews

Title: Toothache and associated factors in Brazilian adults: a cross-sectional population-based study.

Authors:

Mirian Kuhnen (kuhnen@uniplac.net)
Marco A Peres (mperes@ccs.ufsc.br)
Anelise V Masiero (anemasiero@uniplac.net)
Karen G Peres (karengp@ccs.ufsc.br)

Version: 8 Date: 16 December 2008

Author's response to reviews: see over
Dear Dr. Graham,

On behalf of all authors, I am submitting the final version of the manuscript “Toothache and associated factors in Brazilian adults: a cross-sectional population-based study” to your appreciation and to editorial analysis of the **BMC Oral Health**.

The text has been revised according to minor pending formatting required. The changes made were listed bellow and are in bold letter in the final version of the manuscript.

Yours sincerely,

Karen Glazer Peres  
Federal University of Santa Catarina, Brazil.
Minor revisions

Page 5, paragraph 3 sentence 3: the word “was” is not in italic letter. It was corrected.

Page 7, final of line 2: lack of full stop. It was corrected.

Page 10, a last but one line: There is not “comma” between “Mandibular” and “Disorder” words. It was corrected.